ORIGINAL ARTICLE



E-52862—A selective sigma-1 receptor antagonist, in peripheral neuropathic pain: Two randomized, doubleblind, phase 2 studies in patients with chronic postsurgical pain and painful diabetic neuropathy

Rafael Gálvez ¹ Victor Mayoral ²	Jesús Cebrecos ³ 💿 Francisco J. Medel ⁴ 💿
Adelaida Morte ³ Mariano Sust ³	Anna Vaqué ³ 💿 Antonio Montes-Pérez ⁵ 💿
Fernando Neira-Reina ⁶ 💿 Luz Cánova	s ⁷ César Margarit ⁸ Didier Bouhassira ⁹

¹Pain Clinic, Virgen de Las Nieves University Hospital, Granada, Spain ²Bellvitge University Hospital,

L'Hospitalet de Llobregat, Barcelona, Spain

³ESTEVE Pharmaceuticals S.A., Barcelona, Spain

⁴Pain Unit, Vall d'Hebron Research Institute, Barcelona, Spain

⁵Pain Unit, Hospital del Mar Research Institute, Barcelona, Spain

⁶Puerto Real University Hospital, Cádiz, Spain

⁷Pain Clinic, Ourense University Hospital Complex (CHUO), Ourense, Spain

⁸Pain Unit, Alicante University General Hospital, Alicante, Spain

⁹Inserm U987, UVSQ, Paris-Saclay University, Ambroise-Paré Hospital, Boulogne-Billancourt, France

Correspondence

Rafael Gálvez, Pain Clinic, Virgen de las Nieves University Hospital, Granada, Spain.

Email: rafaelgalvez@hotmail.com

Abstract

Background: We report the efficacy and safety of E-52862—a selective, sigma-1 receptor antagonist—from phase 2, randomized, proof-of-concept studies in patients with moderate-to-severe, neuropathic, chronic postsurgical pain (CPSP) and painful diabetic neuropathy (PDN).

Methods: Adult patients (CPSP [N=116]; PDN [N=163]) were randomized at a 1:1 ratio to 4 weeks of treatment with E-52862 (CPSP [n=55]; PDN [n=85]) or placebo (CPSP [n=61]; PDN [n=78]) orally once daily. Pain intensity scores were measured using a numerical pain rating scale from 0 (no pain) to 10 (worst pain imaginable). The primary analysis population comprised patients who received study drug with ≥ 1 baseline and on-treatment observation (full analysis set).

Results: In CPSP, mean baseline average pain was 6.2 for E-52862 vs. 6.5 for placebo. Week 4 mean change from baseline (CFB) for average pain was -1.6 for E-52862 vs. -0.9 for placebo (least squares mean difference [LSMD]: -0.9; p = 0.029). In PDN, mean baseline average pain was 5.3 for E-52862 vs. 5.4 for placebo. Week 4 mean CFB for average pain was -2.2 for E-52862 vs. -2.1 for placebo (LSMD: -0.1; p = 0.766). Treatment-emergent adverse events (TEAEs) were reported in 90.9% of E-52862-treated patients vs. 76.7% of placebo-treated patients in CPSP and 34.1% vs. 26.9% in PDN. Serious TEAEs occurred in CPSP only: E-52862: 5.5%; placebo: 6.7%.

Conclusions: E-52862 demonstrated superior relief of CPSP vs. placebo after 4 weeks. Reductions in pain intensity were seen in PDN with E-52862; high placebo response rates may have prevented differentiation between treatments. E-52862 had acceptable tolerability in both populations.

Significance Statement: These proof-of-concept studies validate the mode of action of E-52862, a selective sigma-1 receptor antagonist. In CPSP, E-52862 resulted

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2024 Esteve Pharmaceuticals, S.A and The Author(s). European Journal of Pain published by John Wiley & Sons Ltd on behalf of European Pain Federation - EFIC *.

in clinically meaningful pain relief. In PDN, reductions in pain intensity were seen with E-52862; high placebo response rates may have prevented differentiation between E-52862 and placebo. These findings are clinically relevant given that neuropathic pain is highly incapacitating, lacking effective treatments and representing a significant unmet medical need, and support further development of sigma-1 receptor antagonists for peripheral neuropathic pain.

1 | INTRODUCTION

Peripheral neuropathic pain represents a major unmet clinical need. It affects approximately 7%–10% of the general population (Bouhassira et al., 2008; van Hecke et al., 2014), but is more prevalent among people who undergo surgery (Rosenberger & Pogatzki-Zahn, 2022) and those with diabetes (Bouhassira et al., 2013; Selvarajah et al., 2019; Tesfaye et al., 2011). Over half of patients do not respond to current treatments (Bouhassira & Attal, 2023; Finnerup et al., 2015; Moisset et al., 2021); management is therefore challenging and often suboptimal, impairing quality of life (Colloca et al., 2017).

Chronic postsurgical pain (CPSP), a common surgical complication, is often underdiagnosed and poorly treated (Giaccari et al., 2021; Montes et al., 2015; Rosenberger & Pogatzki-Zahn, 2022). Risk of CPSP varies based on the type of tissue injury, and the site, extent, duration and type of surgery. Procedures associated with a high risk include knee and hip arthroplasty, thoracotomy, amputation and coronary artery bypass grafting surgery (Charlton et al., 2024; Richebé et al., 2018; Rosenberger & Pogatzki-Zahn, 2022; Schug et al., 2019). CPSP shows neuropathic pain characteristics in up to 80% of cases with certain surgeries (Rosenberger & Pogatzki-Zahn, 2022; Schug et al., 2019).

Prevalence of painful diabetic neuropathy (PDN) among patients with diabetes ranges from 8% to 35%, and PDN impacts quality of life (Abbott et al., 2011; Gylfadottir et al., 2020; Risson et al., 2017; Staehelin, 2023; Syed et al., 2023; Ziegler et al., 2022). Treatment of PDN is challenging, and it remains underdiagnosed and undertreated (Jensen et al., 2021; Sloan et al., 2021; Syed et al., 2023).

The calcium ion-sensing chaperone protein sigma-1 receptor (S1R) regulates several processes, including intracellular signalling cascades related to noxious stimuli and nociception (Ruiz-Cantero et al., 2021). S1R is expressed in areas important for pain control in the central and peripheral nervous systems and physically interacts with several membrane proteins including *N*-methyl-*D*-aspartate receptors; in S1R-knockout mice, neuropathic pain-like behaviours were attenuated (Rabiner et al., 2022). S1R emerged as an analgesic target over the last decade, with a potential role in modulating antinociception (Abadias et al., 2013). E-52862 is a novel, selective, S1R antagonist (S1RA) that reduces nerve injury-evoked activity of S1Rmodulated receptors involved in central (spinal) sensitization and pain hypersensitivity (Ruiz-Cantero et al., 2021). In animal models of neuropathic pain, E-52862 crossed the blood-brain barrier, binding to S1R in the central nervous system, demonstrating analgesic activity for neuropathic pain (Bura et al., 2013; Nieto et al., 2012; Romero et al., 2012; Zamanillo et al., 2013). Pharmacokinetic inhuman studies showed E-52862 has rapid, dose-dependent absorption (Abadias et al., 2013), with a monotonic relationship between plasma drug levels and brain S1R occupancy (Rabiner et al., 2022), allowing for once-daily dosing. In phase 1 studies with healthy volunteers, multiple E-52862 doses up to 400 mg daily had acceptable safety, tolerability, pharmacokinetic, and pharmacodynamic profiles (Abadias et al., 2013; Rabiner et al., 2022; Täubel et al., 2015), with no serious adverse events (AEs) or critical risks (Abadias et al., 2013; Rabiner et al., 2022; Täubel et al., 2015). Daily 400-mg dosing was therefore chosen for further study.

E-52862 was investigated for oxaliplatin-induced peripheral neuropathy in patients with colorectal cancer receiving the folinic acid, fluorouracil, and oxaliplatin (FOLFOX) chemotherapy regimen in a randomized, multicentre, double-blind, placebo-controlled, phase 2 trial (Bruna et al., 2018). E-52862 had an acceptable safety profile with evidence of a potential neuroprotective role for chronic cumulative oxaliplatin-induced peripheral neuropathy (Bruna et al., 2018).

We report the analgesic efficacy, safety and tolerability of E-52862 400 mg once daily (QD) from two phase 2, randomized, proof-of-concept studies in patients with moderate-to-severe CPSP and moderate-to-severe PDN.

2 | METHODS

2.1 | Study design and participants

The CPSP Study (SIGM-205, EudraCT Number: 2012– 000402-30) and PDN Study (SIGM-204, EudraCT Number: 2012–000400-14) were randomized, doubleblind, placebo-controlled, phase 2, proof-of-concept studies in patients with moderate-to-severe CPSP and PDN, respectively. The CPSP Study was conducted at 15 centres in Spain, and the PDN Study was conducted at 26 centres across Spain and Romania. Both studies were ≤9 weeks in duration, including a screening period (up to 3 weeks), a 7-day run-in period, a 28-day treatment period and a 7-day follow-up period (Figure 1).

Eligible patients were aged ≥ 18 years with the following: neuropathic CPSP caused by a surgical intervention that occurred \geq 3 months before enrolment (CPSP Study), or type 1 or 2 diabetes and PDN that had been present for \geq 6 months but <5 years, beginning in the feet with relatively symmetrical onset (PDN Study). Average daily pain score using a numerical pain rating scale (NPRS) had to be ≥ 4 at screening and during the run-in period, with ≥ 5 assessments recorded. The neuropathic pain component was identified using the Douleur Neuropathique en 4 Questions (DN4; DN4 score \geq 4) tool in the CPSP Study (Bouhassira et al., 2005). In the PDN Study, the Michigan Neuropathy Screening Instrument (MNSI) was used to assess peripheral neuropathy (MNSI part B score \geq 3) and a NPRS to assess pain intensity (Feldman et al., 1994). Exclusion criteria included any other active medical condition that could interfere with study assessments or compromise patient safety, severe pain from other causes, a clinically relevant laboratory abnormality or use of prohibited medication. Full inclusion and exclusion criteria are listed in the Data S1.

Both studies were conducted in accordance with Good Clinical Practice, the Declaration of Helsinki and all other applicable standards, and all patients provided written informed consent. The CPSP Study protocol was approved by the Spanish Agency of Medicines and Medical Devices (AEMPS; 18 June 2012) and the Reference Ethics Committee, Hospital Clínic de Barcelona, Barcelona (8 June 2012). The PDN Study protocol was approved by the Reference Ethics Committee, Hospital Clínico San Carlos, Madrid (7 June 2012), the AEMPS (14 June 2012), the Romanian National Bioethics Committee for Medicines and Medical Devices (10 July 2014) and the National Agency for Medicines and Medical Devices of Romania (ANM; 11 July 2014).

2.2 | Procedures

Within each trial, eligible patients were randomized at a 1:1 ratio to receive 4 weeks of treatment (until Day 28) with E-52862 or placebo (Figure 1). The randomization methods are described in detail in the Data S1. E-52862 400 mg and placebo QD were given orally in the morning under fasting conditions. Patients attended clinic visits once a week for assessments. Pain intensity was assessed daily through completion of diary cards from the start of the run-in period (Day -7) to the final follow-up visit on Day 35 in both studies. Scores for pain intensity were measured using a NPRS from 0 (no pain) to 10 (worst pain imaginable) and recorded in the diary cards. Allodynia and hyperalgesia scores were measured using a visual analogue scale (VAS) from 0 to 100 mm after applying a stimulus (brushing for allodynia, pinprick for hyperalgesia); higher scores indicated greater pain severity.

In both studies, only medications that did not interfere with the interpretation of study results and those medically indicated for the treatment of AEs were permitted during the study. Patients continued with their chronic medications if they had been stable in the previous month (or last 3 months for diabetes treatment). In both studies, permitted rescue medications were paracetamol (1000 mg orally every 6h) and metamizole (up to 2g orally every 8h). If at any time the investigator considered that the patient required any other analgesic treatment in addition to the study medication or the permitted rescue medication, the patient was discontinued from the study to allow appropriate treatment.

2.3 Outcomes

The CPSP and PDN studies were both exploratory phase 2 studies and, therefore, no formal primary or secondary endpoints were defined. The following efficacy assessments were conducted in both studies: mean pain intensity (24-h average and worst pain) in the previous



FIGURE 1 Study design for the CPSP and PDN studies. CPSP, chronic postsurgical pain; D, day; PDN, painful diabetic neuropathy; QD, once daily; R, randomization.

7 days using a NPRS; mean allodynia and hyperalgesia using a VAS; short-form Brief Pain Inventory (SF-BPI) (Cleeland, 2009); short-form McGill Pain Questionnaire (SF-MPQ) (Melzack, 1987); Patient Global Impression of Change (PGIC); use of rescue medication; and change in Neuropathic Pain Symptom Inventory (NPSI) score from baseline. Pre-specified subgroup analyses were conducted in both studies. Safety assessments included AEs (identified from spontaneous reporting by patients and from asking patients non-leading questions), vital signs and laboratory parameter abnormalities. A full list of outcomes and their definitions are included in the Data S1.

2.4 | Statistical analysis

As the CPSP and PDN studies were both exploratory, sample sizes were not formally calculated, but were estimated according to clinical criteria. A total of 120 patients were planned to be enrolled in each study.

Efficacy analyses were performed in the full analysis set (FAS) and per-protocol analysis set (PPAS) for both studies. Efficacy data reported in this manuscript are primarily based on the FAS, unless stated otherwise. The FAS comprised all randomized patients who took study medication and provided at least one valid baseline and on-treatment observation for efficacy variables. The PPAS comprised a subset of FAS patients who had no major protocol deviations and were on treatment for \geq 7 days. Safety analyses were performed in the safety analysis set, defined as all randomized patients who received at least one dose of study medication.

Time-specific changes from baseline pain intensity scores were analysed using an analysis of variance model. Fixed effects included factors for treatment and centre, and the baseline value of the variable was used as a covariate. Ninety-five percent confidence intervals (CIs) for the least squares (LS) mean of the treatment difference were derived for each time point. A Mantel– Haenszel chi-square test adjusted for study centre was used to analyse the proportions of patients with reductions from baseline of \geq 30% and \geq 50% from baseline in the 24-h average pain score in the previous 7 days using a NPRS, and the proportions of patients needing rescue medication.

2.5 | Role of the funding source

Employees of the sponsor, ESTEVE Pharmaceuticals, participated in the study designs and conduct, data review and interpretation, drafting of the article and decision to submit for publication. Site management, data collection and statistical analyses were conducted by independent contract research organizations (CPSP Study: RPS Strategic Solutions, a division of PRA Health Sciences [acquired by ICON]; PDN Study: Premier Research).

3 | RESULTS

3.1 | Patients

In the CPSP Study, 116 patients were included, with 55 patients randomized to E-52862 and 61 to placebo (FAS). All patients in the E-52862 group and 60 patients in the placebo group (98.4%) received treatment. In total, 16 patients (11 in the E-52862 group and 5 in the placebo group) withdrew from the study due to AEs, withdrawal of consent or for other reasons (Figure 2a). In the PDN Study, 163 patients were included, with 85 patients randomized to E-52862 and 78 to placebo (FAS). All patients in both groups received treatment. Three patients (two in the E-52862 group and one in the placebo group) withdrew due to an AE, lack of efficacy or withdrawal of consent (Figure 2b).

Patients enrolled in the CPSP Study were younger on average than those in the PDN Study; median ages were 53 years (range: 20–81) and 61 years (range: 23–75 years), respectively (Table 1). The CPSP Study also enrolled a higher proportion of female patients (68.4%) relative to the PDN Study (51.5%). Almost all patients were white (CPSP Study: 96.5%; PDN Study: 100%). Mean average and worst pain scores at baseline were higher in the CPSP Study than in the PDN Study. Additionally, patients in both studies had high NPSI scores at baseline, although patients in the CPSP Study had higher mean baseline scores than patients in the PDN Study for NPSI total pain intensity score and all NPSI dimensional scores, with the exception of superficial spontaneous pain, and paresthesia/dysesthesia, which were similar across studies.

Key baseline characteristics were generally balanced between the E-52862 and placebo arms in both studies. In the CPSP Study, the average and worst NPRS scores and DN4 questionnaire scores at baseline were similar in the E-52862 and placebo groups (average NPRS: 6.2 and 6.5, respectively; worst NPRS: 8.6 and 8.4, respectively; DN4 questionnaire score: 6.9 and 7.1, respectively). Average time since surgery was similar between groups (3.3 and 3.9 years, respectively). In the PDN Study, most patients in both the E-52862 and placebo groups were enrolled in Romania and had type 2 diabetes. Across the treatment groups, mean MNSI (part B) score was 5.0–5.1, and the duration of PDN was 2.2–2.3 years. Patients in the E-52862



FIGURE 2 Patient disposition in the (a) CPSP Study and (b) PDN Study. CPSP, chronic postsurgical pain; PDN, painful diabetic neuropathy.

group had a longer mean duration of diabetes than those in the placebo group (13.0 and 10.3 years, respectively) and a slightly lower worst NPRS score at baseline (6.6 vs. 7.1, respectively). In both CPSP and PDN studies, most patients had received prior neuropathic pain treatment (88.2%–94.9% across groups); however, the types of medication used were different in the CPSP and PDN studies (Table 1; Tables S3 and S4).

3.2 | Efficacy

3.2.1 | CPSP study

In the CPSP Study, E-52862 was associated with reductions in both 24-h average pain intensity and worst pain intensity compared with placebo (Figure 3a, Figure 3b and Table 2). Mean average pain score reduced from 6.2 at

TABLE 1 Baseline demographics and clinical characteristics in the CPSP and PDN studies (full analysis set unless indicated otherwise).

	CPSP Study		PDN Study	
	E-52862 n=55	Placebo $n = 59$	E-52862 n=85	Placebo $n = 78$
Median age, years (range)	51.0 (32-81)	55.0 (20-74)	61.0 (23-75)	60.5 (24–74)
Sex, <i>n</i> (%)				
Female	36 (65.5%)	42 (71.2%)	42 (49.4%)	42 (53.8%)
Male	19 (34.5%)	17 (28.8%)	43 (50.6%)	36 (46.2%)
Race, <i>n</i> (%)				
White	51 (92.7%)	59 (100%)	85 (100%)	78 (100%)
Other	4 (7.3%)	0	0	0
Mean weight, kg (SD)	78.0 (15.4) ^a	74.7 (15.6) ^a	85.8 (16.2)	87.3 (16.6)
Country of enrolment, n (%)				
Spain	55 (100%)	59 (100%)	16 (18.8%)	14 (17.9%)
Romania	0	0	69 (81.2%)	64 (82.1%)
Mean average NPRS score at baseline (SD)	6.2 (1.6)	6.5 (1.5)	5.3 (1.7)	5.4 (1.6)
Mean worst NPRS score at baseline (SD)	8.6 (1.2)	8.4 (1.0)	6.6 (1.5)	7.1 (1.5)
Mean NPSI absolute value at baseline (SD)				
Superficial spontaneous pain	5.0 (3.6) ^b	5.5 (3.3) ^b	5.2 (2.6) ^c	5.0 (2.9) ^c
Deep spontaneous pain	10.8 (6.3) ^b	12.4 (4.7) ^b	4.1 (2.6) ^c	4.3 (2.5) ^c
Paroxysmal pain	12.0 (4.3) ^b	11.5 (5.2) ^b	4.3 (2.8) ^c	4.3 (2.5) ^c
Evoked pain	19.6 (6.6) ^b	19.0 (6.2) ^b	4.0 (2.6) ^c	4.5 (2.4) ^c
Paresthesia/dysesthesia	12.1 (4.6) ^b	12.1 (4.6) ^b	6.0 (2.2) ^c	6.1 (2.4) ^c
Total pain intensity score	59.4 (16.1) ^d	60.6 (14.9) ^d	45.8 (19.3) ^d	47.9 (18.2) ^d
Patients who received any prior neuropathic pain treatment, n (%)	49 (89.1%)	56 (94.9%)	75 (88.2%)	69 (88.5%)
Prior neuropathic pain treatment reported in $\geq 8\%$ of patients, $n(\%)$				
Amitriptyline	3 (5.5%)	6 (10.2%)	-	-
Capsaicin	10 (18.2%)	15 (25.4%)	-	-
Dexketoprofen	5 (9.1%)	2 (3.4%)	-	-
Diclofenac	5 (9.1%)	2 (3.4%)	-	-
Duloxetine	12 (21.8%)	11 (18.6%)	-	-
Fentanyl	2 (3.6%)	5 (8.5%)	-	-
Gabapentin	9 (16.4%)	12 (20.3%)	-	-
Ibuprofen	12 (21.8%)	9 (15.3%)	-	-
Lidocaine	17 (30.9%)	20 (33.9%)	-	-
Metamizole	18 (32.7%)	20 (33.9%)	-	-
Paracetamol	24 (43.6%)	30 (50.8%)	-	-
Paracetamol, tramadol hydrochloride	6 (10.9%)	4 (6.8%)	-	-
Pregabalin	27 (49.1%)	27 (45.8%)	-	-
Tapentadol	3 (5.5%)	5 (8.5%)	-	-
Tramadol	14 (25.5%)	15 (25.4%)	-	-

TAE

ABLE 1 (Continued)				European Journal of Pain
	CPSP Study	CPSP Study		
	E-52862 n = 55	Placebo $n = 59$	E-52862 n=85	Placebo n=78
Prior treatment for PDN reported in $\geq 8\%$ of patients, n (%)				
Antidepressants, other	-	-	7 (8.2%)	1 (1.3%)
Other analgesics and antipyretics ^e	-	-	18 (21.2%)	21 (26.9%)
Various alimentary tract and metabolism products	-	-	29 (34.1%)	27 (34.6%)
Vitamin combinations	-	-	44 (51.8%)	41 (52.6%)
Vitamin B1 and B6	-	-	15 (17.6%)	11 (14.1%)
CPSP Study-specific characteristics				
Mean DN4 questionnaire score (SD)	6.9 (1.5)	7.1 (1.4)	-	-
Site of surgery, <i>n</i> (%)				
Abdomen	3 (5.5%)	8 (13.6%)	-	-
Breast	2 (3.6%)	3 (5.1%)	-	-
Gynaecological	1 (1.8%)	0	-	-
Inguinal	10 (18.2%)	12 (20.3%)	-	-
Lower limb	12 (21.8%)	10 (16.9%)	-	-
Upper limb	11 (20.0%)	11 (18.6%)	-	-
Spinal	11 (20.0%)	9 (15.3%)	-	-
Thorax	5 (9.1%)	6 (10.2%)	-	-
Mean time since surgery, years (SD)	$3.3(3.7)^{f}$	$3.9(3.9)^{f}$	-	-
PDN Study-specific characteristics				
Diabetes type, <i>n</i> (%)				
1	-	-	12 (14.1)	11 (14.1)
2	-	-	73 (85.9)	67 (85.9)
Mean duration of diabetes, years (SD)	-	-	13.0 (9.0)	10.3 (7.5)
Mean duration of PDN, years (SD)	_	_	2.3 (1.3)	2.2 (1.4)
Mean HbA1c level, % (SD)	-	-	7.6 (1.3)	7.3 (1.0)

Abbreviations: CPSP, chronic postsurgical pain; DN4, Douleur Neuropathique en 4 Questions; HbA1C, glycosylated haemoglobin; MNSI, Michigan Neuropathy Screening Instrument; NPRS, numerical pain rating scale; NPSI, Neuropathic Pain Symptom Inventory; PDN, painful diabetic neuropathy; SD, standard deviation; -, not applicable/not reported.

^aData are for the safety analysis set: E-52862 (n = 55) and placebo (n = 60).

^bNPSI dimension score ranges: superficial spontaneous pain, 0–10; deep spontaneous pain, 0–20; paroxysmal pain, 0–20; evoked pain, 0–30; paresthesia/ dysesthesia, 0-20.

^cNPSI dimension score ranges: superficial spontaneous pain, 0–10; deep spontaneous pain, 0–10; paroxysmal pain, 0–10; evoked pain, 0–10; paresthesia/ dysesthesia, 0-10.

^dNPSI total pain intensity score ranges: 0–100.

Mean MNSI (part B) score (SD)

^eDrugs included within this classification were pregabalin and gabapentin.

^fData have been converted from days into years.

baseline to 4.4 at Week 4 with E-52862, and from 6.5 to 5.5 with placebo; difference in LS mean change from baseline (CFB) [95% CI]: -0.9 [-1.7, -0.1]; p = 0.029. Mean worst pain score reduced from 8.6 at baseline to 6.5 at Week 4 with E-52862, and from 8.4 to 7.4 with placebo; difference in LS mean CFB [95% CI]: -1.0 [-1.9, -0.1]; p=0.035). Pain scores remained below pre-treatment values in the off-treatment window (Days 29-36) but were not significantly different between treatment groups.

5.1 (0.9)

5.0(1.1)

A clinically significant response to E-52862 was observed in the subgroup of patients with chronic pain following non-spinal surgery, with statistically significant



FIGURE 3 Overall mean 24-h pain intensity by NPRS over the previous 7 days in the CPSP Study (full analysis set): (a) Overall average pain; (b) Overall worst pain; (c) Average pain in patients with non-spinal surgery; and (d) Worst pain in patients with non-spinal surgery. Corresponding data for patients with spinal surgery are reported in Figure S1. CI, confidence interval; CPSP, chronic postsurgical pain; LS, least squares; NPRS, numerical pain rating scale; SD, standard deviation.

differences in the LS mean CFB in average pain (-1.2)[-2.1, -0.4]; p = 0.007) and worst pain (-1.4 [-2.4, -0.4]; p=0.006) at Week 4 for the comparison of E-52862 vs. placebo (Figure 3c and Figure 3d). In the subgroup of patients with pain following spinal surgery, there was no significant between-group difference in the LS mean CFB for average or worst pain. Further pre-specified subgroup analyses by pain intensity at baseline, prior neuropathic pain treatment status, time since surgery, and sex are reported in Table 2, and by age and pain localization in Table S5.

The proportion of patients achieving at least a 30% or 50% reduction in mean pain intensity from baseline to Week 4 with E-52862 was approximately double compared with that for placebo: 34.5% vs. 15.3% for $\ge 30\%$ reduction (p=0.004) and 16.4% vs. 8.5% for \geq 50% reduction (p=0.122), respectively. Mean CFB to Week 4 in SF-MPO scores were similar in the E-52862 and placebo groups. Mean pain reduction from baseline to Week 4 measured

by the SF-BPI was numerically greater with E-52862 vs. placebo (current pain severity: p=0.107; pain interference: p=0.437). The mean CFB to Week 4 in NPSI total pain intensity score was numerically greater for E-52862 than for placebo (LS mean difference, -8.7 [-18.5, 1.1], p=0.083). For the PGIC, a numerically higher proportion of patients selected the "very much improved" or "much improved" categories for global impression of change at Week 4 with E-52862 (37.0%) vs. the placebo group (23.2%). Overall, there was no significant difference in PGIC response rates between E-52862 and placebo (p=0.311, Wilcoxon rank sum) (Table S6). A numerically greater proportion of patients in the placebo group (82.6%) compared with the E-52862 group (65.9%) received rescue medication (p = 0.190; PPAS data) across all on-treatment study visits and the off-treatment visit at Day 35. Mean CFB to Days 7, 14, 21 and 28 in allodynia and hyperalgesia scores are shown in Table S7. Mean NPSI domain and total pain intensity scores at Days 7, 14, 21, 28 and 35 are shown in Figure S2.

TABLE 2 Summary of efficacy outcomes in the CPSP Study (full analysis set unless indicated otherwise).

	E-52862 n=55	Placebo n = 59	Difference in least squares mean (95% CI)	p value	Relative risk (95% CI)
Mean change in average NPRS score	e in the previous 7	days from baseline	e to Week 4 (SD)	-	
All patients	-1.6(1.9)	-0.9 (1.8)	-0.9 (-1.7, -0.1)	0.029	_
Surgery location					
Non-spinal surgery location	n=44 -1.9 (2.0)	n = 50 -0.8 (1.7)	-1.2 (-2.1, -0.4)	0.007	_
Spinal surgery location	n=10 -0.6 (1.3)	n=9 -1.2 (2.2)	-0.2 (-2.1, 1.6)	0.786	_
Pain intensity at baseline					
Moderate $(\ge 4 \text{ to } \le 6)^a$	n = 24 -1.2 (1.6)	n = 17 -0.8 (1.8)	-0.4 (-1.9, 1.1)	0.602	_
Severe (>6) ^a	n = 17 -2.4 (2.6)	n = 28 -1.3 (2.0)	-1.4 (-3.1, 0.3)	0.097	_
Prior neuropathic pain treatment					
Naïve ^a	n = 10 -1.5 (1.9)	n = 12 -0.9 (2.0)	0.8 (-2.5, 4.1)	0.602	_
Non-naïve ^{a,b}	n = 31 -1.7 (2.2)	n = 33 -1.2 (1.9)	-0.9 (-2.1, 0.3)	0.147	_
Time since surgery					
$\leq 400 \text{days}^{a}$	n = 11 -1.4 (1.3)	n = 10 -0.6 (1.7)	0.4 (-1.7, 2.4)	0.699	_
>400 days ^a	n = 30 -1.8 (2.4)	n = 35 -1.2 (2.0)	0.1 (-1.2, 1.4)	0.890	_
Sex					
Female ^a	n = 36 -2.2 (2.3)	n = 42 -1.1 (1.8)	-1.3 (-2.6, 0.1)	0.065	_
Male ^a	n=19 -0.9 (1.4)	n = 17 -1.2 (2.1)	0.6 (-1.3, 2.5)	0.544	-
Mean change in worst NPRS score in	n the previous 7 da	iys from baseline t	o Week 4 (SD)		
All patients	-2.0 (2.3)	-1.0(2.1)	-1.0 (-1.9, -0.1)	0.035	-
Surgery location					
Non-spinal surgery location	n = 44 -2.3 (2.3)	n = 50 -0.9 (1.7)	-1.4 (-2.4, -0.4)	0.006	_
Spinal surgery location	n=10 7.5 (2.0)	n=9 6.3 (2.8)	0.3 (-2.0, 2.6)	0.802	-
Pain intensity at baseline					
Moderate $(\geq 4 \text{ to } \leq 6)^a$	n = 24 -1.7 (2.0)	n = 17 -1.4 (2.8)	-0.1 (-2.1, 2.0)	0.962	-
Severe (>6) ^a	n=17 -2.4 (3.0)	n = 28 -1.2 (1.9)	-1.1 (-3.0, 0.7)	0.208	_
Sex					
Female ^a	n = 36 -2.6 (2.7)	n = 42 -1.4 (2.2)	-0.7 (-2.5, 1.0)	0.400	-
Male ^a	n=19 -1.0 (1.4)	n = 17 -1.1 (2.5)	0.7 (-1.6, 2.9)	0.524	_

(Continues)

10 of 20

TABLE 2 (Continued)

	E-52862 n=55	Placebo $n = 59$	Difference in least squares mean (95% CI)	p value	Relative risk (95% CI)
Patients with a \geq 30% or \geq 50% reduction	n in average NPRS	score in the previo	ous 7 days from baseline to We	ek 4, n (%)	
≥30%	19 (34.5%)	9 (15.3%)	-	0.004	2.7 (1.3, 5.4)
≥50%	9 (16.4%)	5 (8.5%)	-	0.122	2.2 (0.8, 6.3)
Median time to onset of sustained therapeutic improvement (≥1 point reduction in mean NPRS score from baseline, with a ≥30% reduction in average NPRS score from baseline to Week 4), days (95% CI)	18.0 (8.0, NC)	28.0 (8.0, NC)	-	-	-
Mean SF-MPQ change from baseline to	o Week 4 (SD)				
Sensory total score	-4.6 (9.2)	-5.2 (8.8)	0.9 (-2.3, 4.2)	0.579	-
Affective total score	-1.8 (3.8)	-2.2 (3.7)	0.4 (-0.9, 1.6)	0.589	-
Overall total score	-6.4 (12.2)	-7.3 (11.6)	1.2 (-3.1, 5.6)	0.574	-
Visual analogue scale, cm	-2.0 (3.1)	-1.4 (2.6)	-0.8 (1.8, 0.2)	0.135	-
Mean SF-BPI change from baseline to	Week 4 (SD)				
Items of interest for CPSP Study					
Current pain severity score	-2.1 (3.1)	-1.6 (2.8)	-0.9 (-1.9, 0.2)	0.107	-
Pain interference score	-2.2 (3.2)	-1.7 (3.3)	-0.5(-1.7, 0.7)	0.437	-
Mean NPSI change from baseline to W	eek 4 (SD)				
Superficial spontaneous pain	-1.6 (3.6)	-1.8 (3.5)	-0.1 (-1.2, 1.1)	0.872	-
Deep spontaneous pain	-3.0(7.7)	-2.3 (6.9)	-2.6(-4.9, -0.3)	0.027	-
Faroxysmal pain	-3.7(6.8) -6.0(8.9)	-3.2(6.7) -3.7(7.6)	-0.3(-2.7, 2.1) -2.7(-5.7, 0.3)	0.781	_
Paresthesia/dysesthesia	-4.9(6.9)	-4.0(5.4)	-1.2(-3.4, 1.1)	0.301	_
Total pain intensity score	-19.7 (28.4)	-14.5 (23.7)	-8.7 (-18.5, 1.1)	0.083	-
Patient global impression of change at	Week 4, %			0.311	
Patients in "very much improved" or "much improved" categories	20 (37.0)	13 (23.2)	-		-
Patients in five remaining categories from "minimally improved" to "very much worse"	34 (63.0)	43 (76.8)	-		-
Patients receiving rescue medication, $n (\%)^{a}$	27 (65.9)	38 (82.6)	-	0.190	0.9 (0.7, 1.1)

Abbreviations: CI, confidence interval; CPSP, chronic postsurgical pain; NC, not calculable; NPRS, numerical pain rating scale; NPSI, Neuropathic Pain Symptom Inventory; SD, standard deviation; SF-BPI, short-form Brief Pain Inventory; SF-MPQ, short-form McGill Pain Questionnaire; –, not applicable/not reported.

^aData are for the per-protocol analysis set; E-52862 (n=41) and placebo (n=46). Baseline NPRS score was missing in 1 patient in the placebo group, therefore subgroup analyses data are shown for n=45 patients in the placebo group of the per-protocol analysis set. Since this was a phase 2 study, some exploratory analyses, such as the subgroup analyses, were performed on the per-protocol population.

^bNon-naïve patients were those with any prior treatment with amitriptyline, capsaicin, duloxetine, gabapentin, lidocaine, opioids, or pregabalin.

3.2.2 | PDN study

In the PDN Study, mean values for 24-h average and worst pain intensity over the previous 7 days were similar with E-52862 and placebo and showed a constant reduction between baseline and Week 4 (Figure 4a,b). Mean average

pain score was 5.3 at baseline and 3.0 at Week 4 with E-52862; and 5.4 at baseline and 3.2 at Week 4 with placebo. Mean worst pain score was 6.6 at baseline and 4.2 at Week 4 with E-52862; and 7.1 at baseline and 4.5 at Week 4 with placebo. No significant difference was observed between groups for the CFB to Week 4 in average pain intensity



FIGURE 4 Mean 24-h average (a) and worst (b) pain intensity by NPRS over the previous 7 days in the PDN Study (full analysis set). CI, confidence interval; LS, least squares; NPRS, numerical pain rating scale; PDN, painful diabetic neuropathy; SD, standard deviation.

(difference in LS mean CFB [95% CI]: -0.1 [-0.6, 0.4]; p=0.766) or worst pain intensity (difference in LS mean CFB [95% CI]: 0.2 [-0.3, 0.7]; p=0.461) (Table 3). In subgroup analyses performed by baseline pain intensity, and by sex (Table 3), age, and HbA1C levels (Table S8), no significant differences in average or worst pain at Week 4 were found between groups.

The proportion of patients with at least a 30% or 50% reduction in average pain score from baseline to Week 4 was numerically higher with E-52862 vs. placebo: 62.4% vs. 59.0% for \geq 30% reduction (*p* = 0.856) and 38.8% vs. 34.6% for ≥50% reduction (p = 0.636) respectively (Table 3). Mean reductions from baseline to Week 4 in SF-MPQ scores appeared greater with placebo than E-52862. Analysis of SF-BPI items 5 (pain on average) and 3 (pain at its worst) showed comparable time profiles to NPRS-measured average and worst pain, respectively (data not shown), and similar mean CFB to Week 4 in the two treatment groups (current pain severity: p = 0.478; pain interference: p = 0.112). For the PGIC, a numerically higher proportion of patients selected the "very much improved" or "much improved" categories for global impression of change at Week 4 with E-52862 (52.4%) vs. placebo (43.4%). There was no difference in CFB to Week 4 in NPSI total pain intensity score between E-52862 and placebo (p = 0.422). Overall, there was no significant difference in PGIC response rates between E-52862 and placebo (p=0.716, Wilcoxon rank sum) (Table S6). The proportion of patients who needed rescue medication was similar in the two treatment groups (12.9% with E-52862 vs. 10.3% with placebo; p = 0.642).Mean CFB to Days 7, 14, 21 and 28 in allodynia and hyperalgesia scores are shown in Table S9. Mean NPSI domain and total pain intensity scores at Days 7, 14, 21, 28 and 35 are shown in Figure S3.

3.3 | Safety

In the CPSP Study, 50 patients (90.9%) in the E-52862 group and 46 (76.7%) patients in the placebo group experienced at least one treatment-emergent adverse event (TEAE) (Table 4). The most commonly reported TEAEs were dizziness, headache, nausea, upper abdominal pain, and vomiting. Most TEAEs were of mild or moderate severity in both treatment groups. More patients had TEAEs that were assessed as related to study treatment in the E-52862 group (65.5%) compared with the placebo group (38.3%). TEAEs led to study discontinuation in 5 patients (9.1%) in the E-52862 group and were considered related to study treatment in 4/5 patients: nausea and sense of oppression; nausea, headache, upper abdominal pain and paraesthesia; vertigo; and depression. TEAEs led to discontinuation in 2 patients (3.3%) in the placebo group and were considered related to study treatment in both patients: vomiting and upper abdominal pain.

In the PDN Study, 29 patients (34.1%) in the E-52862 group and 21 patients (26.9%) in the placebo group experienced at least one TEAE (Table 4). The most commonly reported TEAEs were dizziness, nausea, headache, and anxiety. Most TEAEs were mild in severity and were not considered related to E-52862. One patient in the E-52862 group had a TEAE leading to study discontinuation (urinary tract infection), which was assessed as not related to study treatment.

There were no deaths in either study, and serious TEAEs were only reported in the CPSP Study; there were no serious TEAEs in the PDN Study. In the CPSP Study, serious TEAEs were reported by 3 patients (5.5%) in the E-52862 group and 4 patients (6.7%) in the placebo group, and all had resolved at the time of follow-up. In the E-52862 group, the patients with serious TEAEs

TABLE 3 Summary of efficacy outcomes in the PDN Study (full analysis set).

	E-52862 n=85	Placebo $n = 78$	Difference in least squares mean (95% CI)	p value	Adjusted odds ratio (95% CI)	
Mean change in average NPRS score in	n the previous 7 da	ays from baseline	to Week 4 (SD)			
All patients	-2.2 (1.8)	-2.1 (1.8)	-0.1 (-0.6, 0.4)	0.766	-	
Pain intensity at baseline						
Moderate (>4 to <6)	$n = 43^{a}$ -2.1 (1.7)	$n = 46^{a}$ -2.0 (1.6)	-0.0 (-0.7, 0.7)	0.993	-	
Severe (≥6)	$n = 24^{a}$ -3.2 (1.8)	$n = 20^{a}$ -3.2 (2.2)	-0.1 (-1.4, 1.2)	0.925	_	
Sex						
Female	n = 42 -2.5 (1.8)	n = 42 -2.3 (1.9)	-0.3 (-1.0, 0.4)	0.366	-	
Male	n = 43	n = 36				
	-1.9 (1.7)	-1.8 (1.7)	0.1(-0.6, 0.8)	0.787	-	
Mean change in worst NPRS score in t	the previous 7 days	s from baseline to	Week 4 (SD)			
All patients	-2.3 (1.8)	-2.6 (1.9)	0.2 (-0.3, 0.7)	0.461	-	
Pain intensity at baseline						
Moderate (>4 to <6)	$n = 43^{a}$ -2.4 (1.5)	$n = 46^{a}$ -2.6 (1.8)	0.2 (-0.4, 0.9)	0.472	-	
Severe (≥6)	$n = 24^{a}$ -2.8 (2.2)	$n = 20^{a}$ -3.1 (2.4)	0.3 (-1.0, 1.7)	0.630	-	
Sex						
Female	n = 42 -2.7 (1.8)	n = 42 -2.7 (2.0)	-0.2 (-1.0, 0.6)	0.617	_	
Male	n = 43 -2.0 (1.7)	n = 36 -2.3 (1.9)	0.5 (-0.3, 1.2)	0.197	-	
Patients with a \geq 30% or \geq 50% reductio	n in average NPR	S score in the prev	vious 7 days from baseline to V	Veek 4, n (%)		
≥30%	53 (62.4%)	46 (59.0%)	-	0.856	1.1 (0.5, 2.1)	
≥50%	33 (38.8%)	27 (34.6%)	_	0.636	1.2 (0.6, 2.4)	
Median time to onset of sustained therapeutic improvement (≥ 1 point reduction in mean NPRS score from baseline, with a $\geq 30\%$ reduction in average NPRS score from baseline to Week 4), days (95% CI)	20.0 (14.0, 27.0)	20.0 (14.0, 31.0)	-	-	-	
Mean SF-MPQ change from baseline t	o Week 4 (SD)					
Sensory score	-3.9 (5.2)	-5.1 (6.1)	1.2 (-0.3, 2.6)	0.106	_	
Affective score	-1.0 (2.2)	-1.6 (2.2)	0.5 (-0.1, 1.1)	0.084	_	
Total score	-4.8 (6.7)	-6.8 (7.5)	1.9 (0.1, 3.7)	0.042	_	
Mean SF-BPI change from baseline to Week 4 (SD)						
Items of interest for PDN Study						
Current pain severity score	-1.7 (2.1)	-2.2 (2.1)	0.2 (-0.3, 0.7)	0.478	-	
Pain interference score	-1.6 (1.5)	-1.9 (1.8)	0.3 (-0.1, 0.7)	0.112	-	
Mean NPSI change from baseline to Week 4 (SD)						
Superficial spontaneous pain	-1.8 (2.5)	-1.9 (2.6)	0.2 (-0.3, 0.8)	0.400	_	
Deep spontaneous pain	-1.6 (2.1)	-2.0 (2.3)	0.2 (-0.3, 0.6)	0.465	-	
Paroxysmal pain	-1.7 (2.4)	-1.8 (2.3)	-0.0 (-0.6, 0.5)	0.879	-	
Evoked pain	-1.4(2.0)	-1.9(2.1)	0.3(-0.3, 0.8)	0.322	-	
Total pain intensity score	-2.1(2.3) -16.5(17.0)	-2.0(2.2) -20.4(16.9)	1.7(-2.5, 6.0)	0.135	-	

TABLE 3 (Continued)

	E-52862 n=85	Placebo $n = 78$	Difference in least squares mean (95% CI)	p value	Adjusted odds ratio (95% CI)
Patient global impression of change at Week 4, n (%)				0.716	
Patients in "very much improved" or "much improved" categories	44 (52.4 %)	33 (43.4 %)	-		-
Patients in five remaining categories from "minimally improved" to "very much worse"	40 (47.6 %)	42 (55.3 %)	-		-
Patients receiving rescue medication, <i>n</i> (%)	11 (12.9%)	8 (10.3%)	-	0.642	0.7 (0.2, 2.6)

Abbreviations: CI, confidence interval; NPRS, numerical pain rating scale; NPSI, Neuropathic Pain Symptom Inventory; PDN, painful diabetic neuropathy; SD, standard deviation; SF-BPI, short-form Brief Pain Inventory; SF-MPQ, short-form McGill Pain Questionnaire; -, not applicable/not reported. ^aData from the subgroup of patients with mild baseline pain severity (E52862 n = 18; placebo n = 12) are not included here.

were: one patient with increased blood creatinine, potassium and urea (all of which were assessed as related to study treatment); one patient with infection (assessed as unrelated to study treatment and resulting in discontinuation from the study); and one patient with procedural pain (assessed as unrelated to study treatment). In the placebo group, the patients with serious TEAEs were: one patient with increased alanine aminotransferase and gamma-glutamyl transferase (both assessed as unrelated to study treatment); one patient with vomiting (assessed as related to study treatment and resulting in discontinuation from the study); one patient with gastrointestinal obstruction (assessed as unrelated to study treatment); and one patient with abdominal pain (assessed as related to study treatment and resulting in discontinuation from the study).

4 | DISCUSSION

We report analyses of two phase 2, randomized, proof-ofconcept studies of E-52862, a novel, selective S1RA, in patients with moderate-to-severe CPSP and PDN.

In patients with CPSP, E-52862 demonstrated greater reductions from baseline in average and worst 24-h pain compared with placebo. While the magnitude of pain relief based on changes in average pain intensity were small, the improvement in 30% responder rates with E-52862 vs. placebo, combined with the difficult-to-treat population (patients with moderate-to-severe chronic pain, most of whom had shown an unsatisfactory response to other drugs), suggest a clinically relevant treatment effect. Clinical relevance is further emphasized by the scarcity of evidence for CPSP treatment (Kim et al., 2023); most studies in peripheral neuropathic pain have been conducted in post-herpetic neuralgia or PDN (Attal et al., 2023) and efficacy is typically extrapolated in other peripheral neuropathic pain conditions. Moreover, other peripheral neuropathic pain trials have demonstrated smaller pain score improvements despite longer treatment duration (Aiyer et al., 2017; Markman et al., 2018; Shetty et al., 2024).

Results of the CPSP Study subgroup analyses numerically mirrored the main analysis, although treatment differences were not all statistically significant, perhaps due to small sample sizes and short study duration. Notably, average and worst pain were reduced in patients with non-spinal surgery; a potentially more generalizable result than the spinal surgery subgroup, where the poorer response is likely due to the mixed, rather than solely neuropathic, pain associated with spinal surgery (Bajwa & Haldar, 2015) and potential confounding effects of failed back surgery syndrome (Sebaaly et al., 2018). E-52862 appeared to improve pain scores regardless of baseline pain intensity (2.4-point reduction from baseline pain scores at Week 4 in patients with severe baseline pain), prior neuropathic pain treatment, or time since surgery, but these observations did not reach statistical significance. This is clinically meaningful for chronic pain relief and the refractory population, where the mean time since surgery was 3.6 years and around 75% of patients had been treated unsuccessfully with approved or recommended treatments for peripheral neuropathic pain (e.g. pregabalin, gabapentin, duloxetine, capsaicin, amitriptyline, lidocaine, or opioids). Furthermore, there are few effective treatments for CPSP, with limited clinical evidence for most interventions (Attal et al., 2023; Rosenberger & Pogatzki-Zahn, 2022).

In patients with moderate-to-severe PDN, there was a similar reduction in 24-h average and worst pain intensity from baseline, as well as in NPSI score, in both the E-52862 and placebo groups. Although a placebo effect is not unusual for clinical pain trials, particularly PDN (Dworkin

TABLE 4 Summary of TEAEs in the CPSP and PDN studies (safety analysis set).

	CPSP Study		PDN Study		
	E-52862 n=55	Placebo $n = 60$	E-52862 n=85	Placebo <i>n</i> = 78	
Any TEAE, <i>n</i> (%)	50 (90.9%)	46 (76.7%)	29 (34.1%)	21 (26.9%)	
Mild	43 (78.2%)	41 (68.3%)	22 (25.9%)	13 (16.7%)	
Moderate	25 (45.5%)	20 (33.3%)	6 (7.1%)	8 (10.3%)	
Severe	12 (21.8%)	5 (8.3%)	1 (1.2%)	0	
Any treatment-related TEAE, <i>n</i> (%)	36 (65.5%)	23 (38.3%)	11 (12.9%)	8 (10.3%)	
Any TEAE leading to study discontinuation, <i>n</i> (%)	5 (9.1%)	2 (3.3%)	1 (1.2%)	0	
Any serious TEAEs, <i>n</i> (%)	3 (5.5%)	4 (6.7%)	0	0	
Deaths, $n(\%)$	0	0	0	0	
Most common TEAEs by preferred term (occurring in $\geq 5\%$ of patients in any group in either study), $n(\%)^a$					
Dizziness	24 (43.6%)	15 (25.0%)	7 (8.2%)	5 (6.4%)	
Nausea	18 (32.7%)	4 (6.7%)	8 (9.4%)	0	
Headache	17 (30.9%)	13 (21.7%)	3 (3.5%)	5 (6.4%)	
Upper abdominal pain	14 (25.5%)	7 (11.7%)	2 (2.4%)	1 (1.3%)	
Vomiting	7 (12.7%)	1 (1.7%)	3 (3.5%)	0	
Anxiety	6 (10.9%)	2 (3.3%)	0	5 (6.4%)	
Dyspepsia	6 (10.9%)	1 (1.7%)	2 (2.4%)	1 (1.3%)	
Insomnia	6 (10.9%)	4 (6.7%)	0	1 (1.3%)	
Abdominal discomfort	6 (10.9%)	0	0	0	
Arthralgia	5 (9.1%)	4 (6.7%)	0	1 (1.3%)	
Dry mouth	5 (9.1%)	1 (1.7%)	1 (1.2%)	0	
Back pain	4 (7.3%)	2 (3.3%)	1 (1.2%)	2 (2.6%)	
Hyperhidrosis	4 (7.3%)	0	0	0	
Pain in extremity	4 (7.3%)	5 (8.3%)	3 (3.5%)	1 (1.3%)	
Nasopharyngitis	4 (7.3%)	1 (1.7%)	2 (2.4%)	1 (1.3%)	
Constipation	4 (7.3%)	0	3 (3.5%)	0	
Malaise	3 (5.5%)	3 (5.0%)	2 (2.4%)	1 (1.3%)	
Flatulence	3 (5.5%)	0	0	0	
Neck pain	3 (5.5%)	3 (5.0%)	0	2 (2.6%)	
Somnolence	3 (5.5%)	2 (3.3%)	0	2 (2.6%)	
Diarrhoea	3 (5.5%)	1 (1.7%)	0	0	
Muscle spasms	2 (3.6%)	5 (8.3%)	0	0	
Fatigue	2 (3.6%)	4 (6.7%)	2 (2.4%)	2 (2.6%)	
Asthenia	2 (3.6%)	3 (5.0%)	1 (1.2%)	0	
Toothache	1 (1.8%)	3 (5.0%)	0	0	
Abdominal pain	1 (1.8%)	3 (5.0%)	1 (1.2%)	1 (1.3%)	
Limb discomfort	0	3 (5.0%)	0	0	

Abbreviations: TEAE, treatment-emergent adverse event.

^aData are ordered by decreasing incidence in the E-52862 group in the CPSP Study.

et al., 2010), the magnitude of the placebo response was particularly high in the context of other randomized controlled trials (Colloca, 2019; Frisaldi et al., 2017; Frisaldi et al., 2023; Vase & Wartolowska, 2019). Outside of studyrelated elements (e.g. study duration, choice of naïve patients), several factors may contribute to placebo responses in pain studies, including genetic, psychological and biological characteristics (Vase & Wartolowska, 2019). Faceto-face study visits (Vase et al., 2015), and the older average age of patients in the PDN Study compared with the CPSP Study, may have been contributing factors to the observed effect. Patients' expectation of a therapeutic intervention and its efficacy are key drivers (Colloca, 2019; Frisaldi et al., 2017; Vase & Wartolowska, 2019). Development of more precise placebo controls and blinding procedures, and assessment of patients' expectations in routine practice, are needed to control this effect (Frisaldi et al., 2017; Vase & Wartolowska, 2019). Methodological factors such as these, and more general regional differences in the conduct of the CPSP and PDN studies reported here, may have contributed to the observed difference in treatment effects between the two studies. There were also differences in prior pain medication exposure; use of medications with a neuropathic pain indication or endorsement by clinical practice guidelines was marginal in the PDN Study compared to the CPSP Study.

Development of neuropathic pain interventions acting on new targets is challenging; many monotherapies have only moderate efficacy and/or limiting side effects (Attal et al., 2023; Cavalli et al., 2019; Chincholkar, 2020). For example, despite promising outcomes in animal models, the sodium channel blocker lacosamide did not demonstrate improved pain relief vs. placebo in patients with peripheral neuropathic pain (Carmland et al., 2024). In addition, recent phase 2b studies of the selective angiotensin II type 2 receptor antagonist EMA-401 in patients with post-herpetic neuralgia (EMPHENE) and PDN (EMPADINE) showed a greater numerical reduction in NPRS score with EMA-401 vs. placebo (Rice et al., 2021). Alongside improving monotherapy benefit:risk profiles, multimodal treatment approaches that combine different mechanisms of action with the goal of achieving better pain relief through additive effects using lower drug doses could be an alternative approach (Bates et al., 2019).

It is possible that larger treatment effects may have been observed in both studies reported here if higher doses or a longer duration of active treatment were investigated; this would be consistent with preclinical pharmacokinetic and pharmacodynamic modelling (Abadias et al., 2013; Romero et al., 2012; Täubel et al., 2015).

E-52862 had an acceptable safety profile in CPSP and PDN. Considerably fewer TEAEs were reported in the PDN Study compared with the CPSP Study, possibly due to differences in the study centre locations and potential regional variations in TEAE reporting (Guimarães et al., 2017; Keebler et al., 2020). The higher mean weight of patients in the PDN Study compared with the CPSP Study may also have contributed. Except for one patient who reported serious TEAEs of increased blood creatinine, potassium and urea (all considered related to E-52862), serious TEAEs in the CPSP Study were in accordance with the known safety profile of E-52862. No serious TEAEs were reported in the PDN Study. This is a clear distinction from classical first-line drugs for peripheral neuropathic pain.

Results of the CPSP and PDN studies are complemented by a previous exploratory, randomized, doubleblind, placebo-controlled, phase 2 trial of E-52862 in patients with postherpetic neuralgia (SIGM-203) (EU Clinical Trials Register, 2013). Numerically greater reductions in average and worst pain were seen with E-52862 400 mg QD vs. placebo over 28 days; however, the study was terminated early due to poor recruitment, and no statistical comparisons were performed (EU Clinical Trials Register, 2013).

Patient selection is key for conclusive assessment of study treatments against placebo in neuropathic pain trials. Recent guidelines from the International Association for the Study of Pain emphasize the importance of choosing the correct patient screening questionnaires, and provide a strong recommendation for the use of DN4 and the Leeds Assessment of Neuropathic Symptoms and Signs (LANSS) (Truini et al., 2023). Our use of DN4 for patient selection is, therefore, a validated approach and strength of the CPSP Study. Similarly, the difference in NPSI dimensions and total scores between patients in the CPSP and PDN studies and the lack of sensitivity in the NPSI total pain score at the end of treatment in our studies, similar to previously published studies (Rice et al., 2021), confirms the importance of sensory phenotyping at baseline in clinical trials of neuropathic pain conditions. Some flawed studies in neuropathic pain have shown significant effects on specific symptoms, suggesting potential influence of different neuropathic pain phenotypes. The existence of multiple neuropathic pain phenotypes is now established (Bouhassira et al., 2021; van Velzen et al., 2020), and inadequate classification of these phenotypes has been suggested as one reason for failure within neuropathic pain trials (Baron et al., 2012; Baron et al., 2023; Bouhassira & Attal, 2019; Marchevsky et al., 2021). Although the European Medicines Agency endorses patient stratification based on sensory phenotype in clinical trials, this approach is untested in largescale trials (Attal et al., 2023; Bouhassira & Attal, 2023). MNSI, a validated tool for diagnosing distal peripheral diabetic neuropathy, was used in the PDN Study and,

once diagnosed and if painful, the intensity of pain was assessed using a NPRS scale. A specific neuropathic pain assessment scale (e.g. DN4, LANSS) was not used, which is a potential limitation of the PDN Study.

Both the CPSP and PDN studies were proof-of-concept studies. Limitations include the short study treatment duration, relatively small number of patients (who were predominantly White), and the previously discussed high placebo response rates in the PDN Study, which may not have allowed for differentiation from E-52862. A substantial proportion of patients included in the CPSP Study were a refractory, difficult-to-treat population and although this may limit the generalization of the study results, this also increases the potential clinical relevance of these findings given limited effective treatment options for these patients. In this report, we have presented the analyses previously performed for these two proof-of-concept trials; however, further analyses exploring the effects of modifiers of response are warranted. Of particular interest is a post-hoc analysis of how response to E-52862 may differ by specific sensory phenotype, as defined by NPSI dimensional scores at baseline, using the algorithm developed by Bouhassira and colleagues (Bouhassira et al., 2021). The relatively high frequency of TEAEs observed in the CPSP Study could have resulted in guessing of a patient's treatment assignment and consequently, possible bias in assessing some outcomes. The study design mitigated against this by keeping all patients, investigators and personnel involved in the study blinded to the identity of treatment received by each patient, until the clean study database was locked for analysis. The most reported TEAE in the CPSP Study was dizziness in both treatment groups, suggesting the mitigation measures were successful in minimizing potential bias.

The efficacy, safety and tolerability evidence for E-52862 tends to validate the S1RA mode of action, supporting its further development in peripheral neuropathic pain. Next-generation S1RA compounds are in the preclinical phase of development and anticipated to enter early-phase clinical development soon. To progress the clinical development programme for E-52862 in chronic neuropathic pain, chronic preclinical toxicology data were required in two animal species. The rhesus monkey was used because it has similar exposure and metabolic profiles to humans. However, a suitable second species could not be identified due to the very low exposures of E-52862 in other animal models compared with the exposure in humans. Moreover, the low safety margin (at the established no-observed-adverse-effect level) in the longest chronic rhesus monkey toxicity study (39weeks) would not allow for a dose increase based on the predictions from the pharmacokinetic/pharmacodynamic models in other species. Therefore, the clinical development programme

for E-52862 did not progress to longer duration confirmatory studies in chronic treatments.

5 | CONCLUSIONS

E-52862 provided superior relief of CPSP compared with placebo after 4 weeks of treatment in patients with moderate-to-severe CPSP. For patients with PDN, E-52862 provided reductions in pain intensity; however, high placebo response rates may not have allowed for differentiation of active treatment versus placebo. E-52862 had acceptable tolerability in both patient populations. These studies collectively support continued exploration of the S1RA mode of action in chronic peripheral neuropathic pain.

AUTHOR CONTRIBUTIONS

RG, JC, AM, MS, and AV contributed to the conception or design of the studies. RG, FJM, AM-P, and FNR contributed to data collection. All authors contributed to data analysis and/or interpretation. All authors drafted, edited, or reviewed drafts of the manuscript, approved the final version, and agree to be accountable for the accuracy and integrity of the work.

ACKNOWLEDGEMENTS

We thank all patients and investigators involved in the studies. Scientists employed by the funder participated in the design and conduct of the studies, data review and interpretation, and drafting of the article. All authors had full access to all study data and had final responsibility for the decision to submit for publication. Medical writing support, including developing a draft outline and subsequent drafts in consultation with the authors, collating author comments, copyediting, fact checking and referencing, was provided by Rachel Hubbard, MSc, CMPP, and Daniel Binks, PhD, CMPP, at Aspire Scientific Limited (Bollington, UK).

FUNDING INFORMATION

These studies were funded by ESTEVE Pharmaceuticals S.A. (Barcelona, Spain). Medical writing support was funded by ESTEVE Pharmaceuticals S.A. Scientists employed by the funder participated in the design and conduct of the studies, data review and interpretation, decision to publish and drafting of the article. All authors had full access to all study data and had final responsibility for the decision to submit for publication.

CONFLICT OF INTEREST STATEMENT

RG, VM, FJM, AM-P, FN-R and LC report no conflicts of interest. JC, AM, MS and AV are employees of ESTEVE

Pharmaceuticals S.A. (Barcelona, Spain). DB reports consulting fees from Bayer AG, ESTEVE Pharmaceuticals S.A. and Grünenthal Ltd., and honoraria from Grünenthal Ltd.

DATA AVAILABILITY STATEMENT

ESTEVE Pharmaceuticals S.A. (Barcelona, Spain) will consider requests for de-identified patient-level data and supporting study documents from qualified external researchers. Approval of requests will be at the discretion of ESTEVE Pharmaceuticals S.A. and will depend on the scientific merit of the proposed research and intended use of the data. If approval is granted, a data sharing agreement must be signed, and access to data will be provided only if ESTEVE Pharmaceuticals S.A. has legal authority to provide the data and there are no contradictory requirements relating to regulatory filings or reviews. Proposals should be sent to esteve@esteve.com.

ORCID

Rafael Gálvez https://orcid.org/0000-0002-5384-9919 Victor Mayoral D https://orcid.org/0000-0002-0204-0917 Jesús Cebrecos D https://orcid.org/0000-0003-0238-8638 Francisco J. Medel D https://orcid. org/0000-0002-6208-1681 Adelaida Morte b https://orcid.org/0000-0002-3897-8966 Mariano Sust D https://orcid.org/0000-0003-1908-9271 Anna Vaqué 🗅 https://orcid.org/0000-0001-9799-4619 Antonio Montes-Pérez D https://orcid. org/0000-0003-1346-697X Fernando Neira-Reina 🕩 https://orcid. org/0009-0000-1652-5627 Luz Cánovas 💿 https://orcid.org/0000-0001-9221-4839 César Margarit ^D https://orcid.org/0000-0002-1854-9005 Didier Bouhassira 💿 https://orcid. org/0000-0001-6446-8719

REFERENCES

- Abadias, M., Escriche, M., Vaqué, A., Sust, M., & Encina, G. (2013). Safety, tolerability and pharmacokinetics of single and multiple doses of a novel sigma-1 receptor antagonist in three randomized phase I studies. *British Journal of Clinical Pharmacology*, 75, 103–117. https://doi.org/10.1111/j.1365-2125.2012.04333.x
- Abbott, C. A., Malik, R. A., van Ross, E. R., Kulkarni, J., & Boulton, A. J. (2011). Prevalence and characteristics of painful diabetic neuropathy in a large community-based diabetic population in the U.K. *Diabetes Care*, *34*, 2220–2224. https://doi.org/10.2337/ dc11-1108
- Aiyer, R., Barkin, R. L., & Bhatia, A. (2017). Treatment of neuropathic pain with venlafaxine: A systematic review. *Pain Medicine*, 18, 1999–2012. https://doi.org/10.1093/pm/pnw261
- Attal, N., Bouhassira, D., & Colvin, L. (2023). Advances and challenges in neuropathic pain: A narrative review and future

directions. British Journal of Anaesthesia, 131, 79–92. https://doi.org/10.1016/j.bja.2023.04.021

- Bajwa, S. J. S., & Haldar, R. (2015). Pain management following spinal surgeries: An appraisal of the available options. *Journal of Craniovertebral Junction & Spine*, 6, 105–110. https://doi.org/ 10.4103/0974-8237.161589
- Baron, R., Dickenson, A. H., Calvo, M., Dib-Hajj, S. D., & Bennett, D. L. (2023). Maximizing treatment efficacy through patient stratification in neuropathic pain trials. *Nature Reviews Neurology*, 19, 53–64. https://doi.org/10.1038/s41582-022-00741-7
- Baron, R., Forster, M., & Binder, A. (2012). Subgrouping of patients with neuropathic pain according to pain-related sensory abnormalities: A first step to a stratified treatment approach. *Lancet Neurology*, 11, 999–1005. https://doi.org/10.1016/S1474 -4422(12)70189-8
- Bates, D., Schultheis, B. C., Hanes, M. C., Jolly, S. M., Chakravarthy, K. V., Deer, T. R., Levy, R. M., & Hunter, C. W. (2019). A comprehensive algorithm for management of neuropathic pain. *Pain Medicine*, 20(Supplement 1), S2–S12. https://doi.org/10.1093/ pm/pnz075
- Bouhassira, D. & Attal, N. (2019). The multiple challenges of neuropathic pain. *Neuroscience Letters*, 702, 6–10. https://doi.org/10. 1016/j.neulet.2018.11.054
- Bouhassira, D. & Attal, N. (2023). Personalized treatment of neuropathic pain: Where are we now? *European Journal of Pain*, *27*, 1084–1098. https://doi.org/10.1002/ejp.2120
- Bouhassira, D., Attal, N., Alchaar, H., Boureau, F., Brochet, B., Bruxelle, J., Cunin, G., Fermanian, J., Ginies, P., Grun-Overdyking, A., Jafari-Schluep, H., Lantéri-Minet, M., Laurent, B., Mick, G., Serrie, A., Valade, D., & Vicaut, E. (2005). Comparison of pain syndromes associated with nervous or somatic lesions and development of a new neuropathic pain diagnostic questionnaire (DN4). *Pain*, *114*, 29–36. https://doi.org/ 10.1016/j.pain.2004.12.010
- Bouhassira, D., Branders, S., Attal, N., Fernandes, A. M., Demolle, D., Barbour, J., Ciampi de Andrade, D., & Pereira, A. (2021).
 Stratification of patients based on the Neuropathic Pain Symptom Inventory: Development and validation of a new algorithm. *Pain*, *162*, 1038–1046. https://doi.org/10.1097/j.pain. 000000000002130
- Bouhassira, D., Lanteri-Minet, M., Attal, N., Laurent, B., & Touboul, C. (2008). Prevalence of chronic pain with neuropathic characteristics in the general population. *Pain*, *136*, 380–387. https:// doi.org/10.1016/j.pain.2007.08.013
- Bouhassira, D., Letanoux, M., & Hartemann, A. (2013). Chronic pain with neuropathic characteristics in diabetic patients: A French cross-sectional study. *PLoS One*, *8*, e74195. https://doi.org/10. 1371/journal.pone.0074195
- Bruna, J., Videla, S., Argyriou, A. A., Velasco, R., Villoria, J., Santos, C., Nadal, C., Cavaletti, G., Alberti, P., Briani, C., Kalofonos, H. P., Cortinovis, D., Sust, M., Vaqué, A., Klein, T., & Plata-Salamán, C. (2018). Efficacy of a novel sigma-1 receptor antagonist for oxaliplatin-induced neuropathy: A randomized, double-blind, placebo-controlled phase IIa clinical trial. *Neurotherapeutics*, *15*, 178–189. https://doi.org/10.1007/s13311-017-0572-5
- Bura, A. S., Guegan, T., Zamanillo, D., Vela, J. M., & Maldonado, R. (2013). Operant self-administration of a sigma ligand improves nociceptive and emotional manifestations of neuropathic pain. *European Journal of Pain*, 17, 832–843. https://doi.org/10. 1002/j.1532-2149.2012.00251.x

- Carmland, M. E., Kreutzfeldt, M. D., Holbech, J. V., Brask-Thomsen, P. K., Krøigård, T., Hansen, P. N., Tankisi, H., Jensen, T. S., Bach, F. W., Sindrup, S. H., & Finnerup, N. B. (2024). The effect of lacos-amide in peripheral neuropathic pain: A randomized, double-blind, placebo-controlled, phenotype-stratified trial. *European Journal of Pain*, 28, 105–119. https://doi.org/10.1002/ejp.2165
- Cavalli, E., Mammana, S., Nicoletti, F., Bramanti, P., & Mazzon, E. (2019). The neuropathic pain: An overview of the current treatment and future therapeutic approaches. *International Journal* of *Immunopathology and Pharmacology*, 33, 2058738419838383. https://doi.org/10.1177/2058738419838383
- Charlton, E., Atkins, K. J., Evered, L., Silbert, B., & Scott, D. A. (2024). The long-term incidence of chronic post-surgical pain after coronary artery bypass surgery — a prospective observational study. *European Journal of Pain*, 28, 599–607. https://doi. org/10.1002/ejp.2203
- Chincholkar, M. (2020). Gabapentinoids: Pharmacokinetics, pharmacodynamics and considerations for clinical practice. *British Journal of Pain*, 14, 104–114. https://doi.org/10.1177/20494 63720912496
- Cleeland, C. S. (2009). The brief pain inventory user guide. https:// www.mdanderson.org/documents/Departments-and-Divis ions/Symptom-Research/BPI_UserGuide.pdf
- Colloca, L. (2019). The placebo effect in pain therapies. *Annual Review of Pharmacology and Toxicology*, 59, 191–211. https://doi.org/10.1146/annurev-pharmtox-010818-021542
- Colloca, L., Ludman, T., Bouhassira, D., Baron, R., Dickenson, A. H., Yarnitsky, D., Freeman, R., Truini, A., Attal, N., Finnerup, N. B., Eccleston, C., Kalso, E., Bennett, D. L., Dworkin, R. H., & Raja, S. N. (2017). Neuropathic pain. *Nature Reviews Disease Primers*, *3*, 17002. https://doi.org/10.1038/nrdp.2017.2
- Dworkin, R. H., Turk, D. C., Peirce-Sandner, S., McDermott, M. P., Farrar, J. T., Hertz, S., Katz, N. P., Raja, S. N., & Rappaport, B. A. (2010). Placebo and treatment group responses in postherpetic neuralgia vs. painful diabetic peripheral neuropathy clinical trials in the REPORT database. *Pain*, 150, 12–16. https://doi.org/ 10.1016/j.pain.2010.02.002
- EU Clinical Trials Register (2013). Clinical Trial Results: SIGM-203. https://www.clinicaltrialsregister.eu/ctr-search/trial/2012-000399-41/results
- Feldman, E. L., Stevens, M. J., Thomas, P. K., Brown, M. B., Canal, N., & Greene, D. A. (1994). A practical two-step quantitative clinical and electrophysiological assessment for the diagnosis and staging of diabetic neuropathy. *Diabetes Care*, *17*, 1281– 1289. https://doi.org/10.2337/diacare.17.11.1281
- Finnerup, N. B., Attal, N., Haroutounian, S., McNicol, E., Baron, R., Dworkin, R. H., Gilron, I., Haanpaa, M., Hansson, P., Jensen, T. S., Kamerman, P. R., Lund, K., Moore, A., Raja, S. N., Rice, A. S., Rowbotham, M., Sena, E., Siddall, P., Smith, B. H., & Wallace, M. (2015). Pharmacotherapy for neuropathic pain in adults: A systematic review and meta-analysis. *Lancet Neurology*, *14*, 162–173. https://doi.org/10.1016/S1474-4422(14)70251-0
- Frisaldi, E., Shaibani, A., & Benedetti, F. (2017). Why we should assess patients' expectations in clinical trials. *Pain and Therapy*, 6, 107–110. https://doi.org/10.1007/s40122-017-0071-8
- Frisaldi, E., Vollert, J., Al Sultani, H., Benedetti, F., & Shaibani, A. (2023). Placebo and nocebo responses in painful diabetic neuropathy: Systematic review and meta-analysis. *Pain*, *165*, 29–43. https://doi.org/10.1097/j.pain.0000000000000000

- Giaccari, L. G., Aurilio, C., Coppolino, F., Pace, M. C., Passsavanti, M. B., Pota, V., & Sansone, P. (2021). Capsaicin 8% patch and chronic postsurgical neuropathic pain. *Journal of Personalized Medicine*, 11, 960. https://doi.org/10.3390/jpm11100960
- Guimarães, P. O., Lopes, R. D., Stevens, S. R., Zimerman, A., Wruck, L., James, S. K., Haque, G., Giraldez, R. R. C. V., Alexander, J. H., & Alexander, K. P. (2017). Reporting clinical end points and safety events in an acute coronary syndrome trial: Results with integrated collection. *Journal of the American Heart Association*, 6, e005490. https://doi.org/10.1161/jaha.117.005490
- Gylfadottir, S. S., Itani, M., Kroigard, T., Kristensen, A. G., Christensen, D. H., Nicolaisen, S. K., Karlsson, P., Callaghan, B. C., Bennett, D. L., Andersen, H., Tankisi, H., Nielsen, J. S., Andersen, N. T., Jensen, T. S., Thomsen, R. W., Sindrup, S. H., & Finnerup, N. B. (2020). Diagnosis and prevalence of diabetic polyneuropathy: A cross-sectional study of Danish patients with type 2 diabetes. *European Journal of Neurology*, *27*, 2575– 2585. https://doi.org/10.1111/ene.14469
- Jensen, T. S., Karlsson, P., Gylfadottir, S. S., Andersen, S. T., Bennett, D. L., Tankisi, H., Finnerup, N. B., Terkelsen, A. J., Khan, K., Themistocleous, A. C., Kristensen, A. G., Itani, M., Sindrup, S. H., Andersen, H., Charles, M., Feldman, E. L., & Callaghan, B. C. (2021). Painful and non-painful diabetic neuropathy, diagnostic challenges and implications for future management. *Brain*, 144, 1632–1645. https://doi.org/10.1093/ brain/awab079
- Keebler, D., Teng, E., Chia, J., Galanter, J., Peake, J., & Tuckwell, K. (2020). Regional variations in adverse event reporting rates and ACR responses in placebo/standard-of-care arms of rheumatoid arthritis trials. *Rheumatology (Oxford)*, 59, 3023–3031. https://doi.org/10.1093/rheumatology/keaa043
- Kim, B. R., Yoon, S. H., & Lee, H. J. (2023). Practical strategies for the prevention and management of chronic postsurgical pain. *The Korean Journal of Pain*, 36, 149–162. https://doi.org/10.3344/ kjp.23080
- Marchevsky, E., Guaycochea, S., Bendersky, M., & González, S. (2021). Phenotypes and diagnosis in neuropathic pain. *Multidisciplinary Pain Journal*, 1, 81–91. https://doi.org/10. 20986/mpj.2021.1008/2021
- Markman, J., Resnick, M., Greenberg, S., Katz, N., Yang, R., Scavone, J., Whalen, E., Gregorian, G., Parsons, B., & Knapp, L. (2018).
 Efficacy of pregabalin in post-traumatic peripheral neuropathic pain: A randomized, double-blind, placebo-controlled phase 3 trial. *Journal of Neurology*, 265, 2815–2824. https://doi.org/10. 1007/s00415-018-9063-9
- Melzack, R. (1987). The short-form McGill pain questionnaire. *Pain*, 30, 191–197. https://doi.org/10.1016/0304-3959(87)91074-8
- Moisset, X., Bouhassira, D., & Attal, N. (2021). French guidelines for neuropathic pain: An update and commentary. *Revue Neurologique* (*Paris*), 177, 834–837. https://doi.org/10.1016/j.neurol.2021.07.004
- Montes, A., Roca, G., Sabate, S., Lao, J. I., Navarro, A., Cantillo, J., Canet, J., & GENDOLCAT Study Group (2015). Genetic and clinical factors associated with chronic postsurgical pain after hernia repair, hysterectomy, and thoracotomy: A two-year multicenter cohort study. *Anesthesiology*, 122, 1123–1141. https:// doi.org/10.1097/ALN.00000000000611
- Nieto, F. R., Cendán, C. M., Sánchez-Fernández, C., Cobos, E. J., Entrena, J. M., Tejada, M. A., Zamanillo, D., Vela, J. M., & Baeyens, J. M. (2012). Role of sigma-1 receptors in

paclitaxel-induced neuropathic pain in mice. *The Journal of Pain*, *13*, 1107–1121. https://doi.org/10.1016/j.jpain.2012.08. 006

- Rabiner, E. A., Smith, K., Bennett, C., Rizzo, G., Lewis, Y., Mundin, G., Dooner, H., & Oksche, A. (2022). Pharmacokinetics and brain σ1 receptor occupancy of MR309, a selective σ1 receptor antagonist. *British Journal of Clinical Pharmacology*, 88, 1644– 1654. https://doi.org/10.1111/bcp.14952
- Rice, A. S. C., Dworkin, R. H., Finnerup, N. B., Attal, N., Anand, P., Freeman, R., Piaia, A., Callegari, F., Doerr, C., Mondal, S., Narayanan, N., Ecochard, L., Flossbach, Y., & Pandhi, S. (2021). Efficacy and safety of EMA401 in peripheral neuropathic pain: Results of 2 randomised, double-blind, phase 2 studies in patients with postherpetic neuralgia and painful diabetic neuropathy. *Pain*, *162*, 2578–2589. https://doi.org/10.1097/j.pain. 000000000002252
- Richebé, P., Capdevila, X., & Rivat, C. (2018). Persistent postsurgical pain: Pathophysiology and preventative pharmacologic considerations. *Anesthesiology*, *129*, 590–607. https://doi.org/10.1097/ ALN.00000000002238
- Risson, V., Nallagangula, T. K., & Vasanthaprasad, V. (2017). Incidence and prevalence of painful diabetic neuropathy and postherpetic neuralgia in major 5 European countries, the United States and Japan. *Value in Health*, 20, A547. https://doi. org/10.1016/j.jval.2017.08.845
- Romero, L., Zamanillo, D., Nadal, X., Sánchez-Arroyos, R., Rivera-Arconada, I., Dordal, A., Montero, A., Muro, A., Bura, A., Segalés, C., Laloya, M., Hernández, E., Portillo-Salido, E., Escriche, M., Codony, X., Encina, G., Burgueño, J., Merlos, M., Baeyens, J. M., ... Vela, J. M. (2012). Pharmacological properties of S1RA, a new sigma-1 receptor antagonist that inhibits neuropathic pain and activity-induced spinal sensitization. *British Journal of Pharmacology*, *166*, 2289–2306. https://doi.org/10. 1111/j.1476-5381.2012.01942.x
- Rosenberger, D. C., & Pogatzki-Zahn, E. M. (2022). Chronic postsurgical pain – Update on incidence, risk factors and preventive treatment options. *BJA Education*, 22, 190–196. https://doi.org/ 10.1016/j.bjae.2021.11.008
- Ruiz-Cantero, M. C., Gonzalez-Cano, R., Tejada, M. A., Santos-Caballero, M., Perazzoli, G., Nieto, F. R., & Cobos, E. J. (2021). Sigma-1 receptor: A drug target for the modulation of neuroimmune and neuroglial interactions during chronic pain. *Pharmacological Research*, 163, 105339. https://doi.org/10. 1016/j.phrs.2020.105339
- Schug, S. A., Lavand'homme, P., Barke, A., Korwisi, B., Rief, W., Treede, R. D., & IASP Taskforce for the Classification of Chronic Pain (2019). The IASP classification of chronic pain for *ICD-11*: Chronic postsurgical or posttraumatic pain. *Pain*, *160*, 45–52. https://doi.org/10.1097/j.pain.00000000001413
- Sebaaly, A., Lahoud, M. J., Rizkallah, M., Kreichati, G., & Kharrat, K. (2018). Etiology, evaluation, and treatment of failed back surgery syndrome. *Asian Spine Journal*, *12*, 574–585. https://doi. org/10.4184/asj.2018.12.3.574
- Selvarajah, D., Kar, D., Khunti, K., Davies, M. J., Scott, A. R., Walker, J., & Tesfaye, S. (2019). Diabetic peripheral neuropathy: Advances in diagnosis and strategies for screening and early intervention. *The Lancet Diabetes and Endocrinology*, 7, 938–948. https://doi.org/10.1016/S2213-8587(19)30081-6
- Shetty, A., Delanerolle, G., Cavalini, H., Deng, C., Yang, X., Boyd, A., Fernandez, T., Phiri, P., Bhaskar, A., & Shi, J. Q. (2024). A

systematic review and network meta-analysis of pharmaceutical interventions used to manage chronic pain. *Scientific Reports*, *14*, 1621. https://doi.org/10.1038/s41598-023-49761-3

- Sloan, G., Selvarajah, D., & Tesfaye, S. (2021). Pathogenesis, diagnosis and clinical management of diabetic sensorimotor peripheral neuropathy. *Nature Reviews Endocrinology*, 17, 400–420. https://doi.org/10.1038/s41574-021-00496-z
- Staehelin, J. T. (2023). The pathogenesis of painful diabetic neuropathy and clinical presentation. *Diabetes Research and Clinical Practice*, 206(Supplement 1), 110753. https://doi.org/10.1016/j. diabres.2023.110753
- Syed, O., Jancic, P., & Knezevic, N. N. (2023). A review of recent pharmacological advances in the management of diabetesassociated peripheral neuropathy. *Pharmaceuticals (Basel)*, 16, 801. https://doi.org/10.3390/ph16060801
- Täubel, J., Ferber, G., Lorch, U., Wang, D., Sust, M., & Camm, A. J. (2015). Single doses up to 800 mg of E-52862 do not prolong the QTc interval—a retrospective validation by pharmacokineticpharmacodynamic modelling of electrocardiography data utilising the effects of a meal on QTc to demonstrate ECG assay sensitivity. *PLoS One, 10*, e0136369. https://doi.org/10.1371/ journal.pone.0136369
- Tesfaye, S., Vileikyte, L., Rayman, G., Sindrup, S. H., Perkins, B. A., Baconja, M., Vinik, A. I., Boulton, A. J., & Toronto Expert Panel on Diabetic Neuropathy. (2011). Painful diabetic peripheral neuropathy: Consensus recommendations on diagnosis, assessment and management. *Diabetes/Metabolism Research and Reviews*, 27, 629–638. https://doi.org/10.1002/dmrr.1225
- Truini, A., Aleksovska, K., Anderson, C. C., Attal, N., Baron, R., Bennett, D. L., Bouhassira, D., Cruccu, G., Eisenberg, E., Enax-Krumova, E., Davis, K. D., Di Stefano, G., Finnerup, N. B., Garcia-Larrea, L., Hanafi, I., Haroutounian, S., Karlsson, P., Rakusa, M., Rice, A. S. C., ... Veluchamy, A. (2023). Joint European Academy of Neurology—European Pain Federation—Neuropathic Pain Special Interest Group of the International Association for the Study of Pain guidelines on neuropathic pain assessment. *European Journal of Neurology*, 30, 2177–2196. https://doi.org/10.1111/ene. 15831
- van Hecke, O., Austin, S. K., Khan, R. A., Smith, B. H., & Torrance, N. (2014). Neuropathic pain in the general population: A systematic review of epidemiological studies. *Pain*, 155, 654–662. https://doi.org/10.1016/j.pain.2013.11.013
- van Velzen, M., Dahan, A., & Niesters, M. (2020). Neuropathic pain: Challenges and opportunities. *Frontiers in Pain Research*, 1. https://doi.org/10.3389/fpain.2020.00001
- Vase, L., Vollert, J., Finnerup, N. B., Miao, X., Atkinson, G., Marshall, S., Nemeth, R., Lange, B., Liss, C., Price, D. D., Maier, C., Jensen, T. S., & Segerdahl, M. (2015). Predictors of the placebo analgesia response in randomized controlled trials of chronic pain: A meta-analysis of the individual data from nine industrially sponsored trials. *Pain*, *156*, 1795–1802. https://doi.org/10. 1097/j.pain.00000000000217
- Vase, L., & Wartolowska, K. (2019). Pain, placebo, and test of treatment efficacy: A narrative review. *British Journal of Anaesthesia*, 123, e254–e262. https://doi.org/10.1016/j.bja.2019.01.040
- Zamanillo, D., Romero, L., Merlos, M., & Vela, J. M. (2013). Sigma 1 receptor: A new therapeutic target for pain. *European Journal* of Pharmacology, 716, 78–93. https://doi.org/10.1016/j.ejphar. 2013.01.068



Ziegler, D., Tesfaye, S., Spallone, V., Gurieva, I., Al Kaabi, J., Mankovsky, B., Martinka, E., Radulian, G., Nguyen, K. T., Stirban, A. O., Tankova, T., Varkonyi, T., Freeman, R., Kempler, P., & Boulton, A. J. (2022). Screening, diagnosis and management of diabetic sensorimotor polyneuropathy in clinical practice: International expert consensus recommendations. *Diabetes Research and Clinical Practice*, 186, 109063. https:// doi.org/10.1016/j.diabres.2021.109063

SUPPORTING INFORMATION

Additional supporting information can be found online in the Supporting Information section at the end of this article. How to cite this article: Gálvez, R., Mayoral, V., Cebrecos, J., Medel, F. J., Morte, A., Sust, M., Vaqué, A., Montes-Pérez, A., Neira-Reina, F., Cánovas, L., Margarit, C., & Bouhassira, D. (2025). E-52862—A selective sigma-1 receptor antagonist, in peripheral neuropathic pain: Two randomized, double-blind, phase 2 studies in patients with chronic postsurgical pain and painful diabetic neuropathy. *European Journal of Pain, 29*, e4755. <u>https://doi.org/10.1002/</u> ejp.4755